Author’s response to reviews

Title: Clinical efficacy of implantation of toric intraocular lenses with different incision positions: a comparative study of steep-axis incision and non-steep-axis incision

Authors:

Wenwen He (alicewen@139.com)
Xiangjia Zhu (zhuxiangjia1982@163.com)
Yu Du (joycedu624@163.com)
Jin Yang (jin_er76@hotmail.com)
Yi Lu (luyieent@126.com)

Version: 2 Date: 11 Jul 2017

Author’s response to reviews:

Response:

1. Thank you for providing an Availability of data and materials section in the Declarations. Please ensure that all data to be shared through the link http://dx.doi.org/10.5061/dryad.tk492 is published. As of now the link is not functional.

Thank you for your question. All data files are available from the Dyrad database and the DOI is http://dx.doi.org/10.5061/dryad.01n20

2. Please include an Acknowledgements subheading in your Declarations section. In this section acknowledge anyone who contributed towards the article who does not meet the criteria for authorship including anyone who provided professional writing services or materials. Please note that any sections that are not applicable to your manuscript should still be included with the statement “Not applicable.”

Thank you for your reminding. We have modified it as your requirement.

3. Please provide a List of abbreviations after the Conclusions section. If abbreviations are used in the text, they should be defined in the text at first use and included in this list.

Thank you for your suggestion. We have added the abbreviation list after the conclusion.
4. Please change the Purpose heading in the Abstract to a Background heading. Please also introduce the background heading before the sentence beginning “In virtue of improvements in surgical technology, the purpose of cataract surgery…” found on page four.

Thanks a lot and we have made the change as you suggested.

5. Please remove the response to reviewers from the file inventory, as it is no longer needed at this stage of the editorial process.

Thank you and we have removed it.